Author's response to reviews

Title: Oxygen persufflation as adjunct in liver preservation (OPAL): Study protocol for a randomized controlled trial

Authors:

Thomas Minor (tminor@uni-bonn.de)
Carolin Pütter (Carolin.puetter@uk-essen.de)
Anja Gallinat (a.gallinat@gmx.de)
Claudia Ose (Claudia.Ose@uk-essen.de)
Gernot Kaiser (Gernot.Kaiser@uk-essen.de)
Andre Scherag (Andre.Scherag@uk-essen.de)
Jürgen Treckmann (j.treckmann@gmx.net)
Andreas Paul (Andreas.Paul@uk-essen.de)

Version: 7 Date: 28 October 2011

Author's response to reviews:

as requested figure 1 is now cited in the text
some typographica erors have been corrected